Author's response to reviews

Title: Progressive and deleterious multifocal cerebral infarction in a young kidney transplant recipient due to thrombotic microangiopathy

Authors:

Arash Haghikia (haghikia.arash@mh-hannover.de)
Meike Heeren (heeren.meike@mh-hannover.de)
Bernd Haubitz (haubitz.bernd@mh-hannover.de)
Clemens Bockmeyer (bockmeyer.clemens@mh-hannover.de)
Wilfried Gwinner (gwinner.wilfired@mh-hannover.de)

Version: 3 Date: 22 February 2014

Author's response to reviews: see over
Re: Submission of the manuscript “Progressive and deleterious multifocal cerebral infarction in a young kidney transplant recipient due to thrombotic microangiopathy”

Dear Editor,

Please find enclosed our re-revised manuscript entitled “Progressive and deleterious multifocal cerebral infarction in a young kidney transplant recipient due to thrombotic microangiopathy” with the modifications according to the editorial points.

We hope you find the manuscript now suited for publication in BMC Nephrology.

The manuscript, or part of it, has neither been published nor is currently under consideration for publication by any other journal.

I as the submitting and corresponding author declare that the co-authors have read the manuscript and approved its submission to BMC Nephrology.

Thank you for your time and consideration.

Yours sincerely,

Arash Haghikia, MD
Corresponding author